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Challenges and Opportunities for China Fuel Ethanol Industry
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Abstract: With the recent changes of energy and food processing policies, China embraces the great
opportunity of developing fuel ethanol. According to the national plan released by the office of
Premier Minister, the production of fuel ethanol will be greatly increased in the next few years. Apart
from the feedstock,challenges could be from new plants design,operation efficiency,water and
waste water management, education. Chinese fuel industry is still far behind world level ,mindsets
need to be changed and improved. The author reviews the current status of advanced technologies,
advocates the immediate actions needed to be taken on engineering design and education,and finally
prospects for the near future developments.
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Table 1 Production of Major Grains from China and

their rankings in the world
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Table 2 World Production of Fresh Cassava Root(million tons)

263.177 276.766 276.995 278.754 277.957
=M 144.18 154.9 152.833 155.398 155.962 56.11
DIV 88.276 89.365 91.689 90.274 92.418 33.25
FLT WM 30.538 32.334 32.299 32.908 29.407 10.58
KU 0.261 0.239 0.252 0.252 0.247 0.09

>k . Trade and Markets Division(EST). FAO

JOURNAL OF FOOD SCIENCE AND BIOTECHNOLOGY Vol.38 No.5 2019



BOML S, P ERA OB L ALIE 5 Bk

®3 ENESBEAREFE(BAIW)
Table 3 Fresh Cassava Production in Asian Countries

(million tons)

I TN T T

7 74.348 82.611 87.482
e 30.022 32.358 31.807 36.36
e 23.436 22.906 26.749 30.57
fiTe] 10.210 10.664 10.921 12.48
Eht 1.630 1.514 1.527 1.75
CRHE S 7.933 11.944 13.222 15.11
4 fa) 0.524 0.485 0.509 0.58
E[E 2 2.540 2.682 2.682 3.07
ok 75 . 0.052 0.057 0.063 0.7

KR : Trade and Markets Division (EST). FAO
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Fig. 1 Production of Fresh Cassava Roots in Thailand

(Data from Thailand Cassava Association)
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Fig. 2 Fresh Cassava Production in China Over the Past
Years
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Table 4 Pateut applications and granted from ICM
20170107452 Products produced from distillers corn oil
20140080183 Hybrid separation
20130121891 Selected solids separation
20120051980 Bio—oil recovery systems
20110283602 Bio—oil recovery systems and methods
20110265373 Rotary torrefaction reactor

20090311397 Process for edible protein extraction from corn

germ

20090269817 Pretreatment of grain slurry with alpha —amylase

and a hemicellulase blend prior to liquefaction

20080009048 Liquefaction of starch containing material
20040192896 Method for extracting gluten

9718006 Multi-zoned screening apparatus

9376504 Hybrid separation

9375731 Selected solids separation

9150790 Rotary torrefaction reactor

8986551 Suspended solids separation systems and methods
8192627 Bio—oil recovery methods

7572353 Ethanol distillation process

7520990 Anaerobic wastewater treatment system and method
7297236 Ethanol distillation process

7097762 Modular waste water treatment system
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Table 5 Patent applications from Fluid Quip Co.
L= LR AT
9932246  Pulse cavitation processor and method of using same
9777303  Systems and methods for producing a sugar stream
Apparatus and method for filtering a material from a
8813973
liquid medium
Methods for producing a high protein corn meal from
8778433
a whole stillage byproduct
8753505  Liquid treatment apparatus and method for using same
8714811  Multiple helical vortex baffle
8696192  Multiple helical vortex baffle
8012262 Process for steeping corn and steeping system
therefore
6511005 Bowl centrifuge nozzle
6428591  Pressurized cyclone including agitator assembly
Nozzle for centrifuge rotors and method of removing
6216959
same

5499720  Multiple hydrocyclone assembly
5388708  Multiple hydrocyclone assembly
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Table 6 Comparison of Investment(100mgpy plant) and Some Key Indexes of China and US
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Fig. 3 Schematic Flow Diagram of ICM Ethanol Plant
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Possible Integration Scenario
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